Author’s response to reviews

Title: Associations between cardiorespiratory fitness and weight loss in patients with severe obesity undergoing an intensive lifestyle intervention program

Authors:

Jarle Berge (jarle.berge@siv.no)
Øyvind Støren (Oyvind.storen@usn.no)
Jens Hertel (jehert@siv.no)
Espen Gjevestad (espen.svendsen.gjevestad@phs.no)
Milada Småstuen (milasm@oslomet.no)
Jøran Hjelmesæth (joran.hjelmeseth@siv.no)

Version: 1 Date: 05 Jun 2019

Author’s response to reviews:

Dear editor.

We are glad to hear that our manuscript is potentially acceptable for publication in BMC Endocrine Disorders

1. Font size of the P-values and percentages are increased

2. More information about the figures legends are provide in the manuscript. SEM are now included in the figures.

3. We have now clarified about the consent in “ethical approval and consent to participate” section and in the manuscript.

4. Other files are removed from the attach files.
I also see that your systems do not find one of our references (Duvigneaud, N; Matton, L; Wijndaele, K; Deriemaeker, P; Lefevre, J; et al. Journal of Sports Medicine and Physical Fitness).

The reference can be found here: https://search.proquest.com/docview/202677742/fulltextPDF/B83FDA64D846438EPQ/1?accountid=32827

Sincerely

Jarle Berge